Author's response to reviews

Title: Survival of cancer patients treated with mistletoe extract (Iscador): a systematic literature review

Authors:

Thomas Ostermann (thomaso@uni-wh.de)
Christa Raak (c.raak@gemeinschaftkrankenhaus.de)
Arndt Buessing (ardt.buessing@uni-wh.de)

Version: 3 Date: 30 October 2009

Author's response to reviews: see over
Dear Sir.

Thanks a lot for the information that our manuscript is – in principle – suited for publication. Malinee Laopaiboon’s statement that we should clarify in which studies the strata were found was right. We specified this as follows:

**Iscador versus no extra treatment**

Forty one strata (i.e., localization, stages, lymph nodes etc.) found in 22 studies (Figure 1, Table 1) provided enough data to extract HRs and their standard errors. (…)

Moreover, we adapted Figure 1.

Again, thanks a lot for your kind suggestions!
If there any open questions, do not hesitate to contact me.
With best wishes (in behalf)
Yours

Arndt Büssing